**Creating Opportunities** 



## N.º 22/12 Working paper

# Converging to Convergence: The Role of Human Capital

Amparo Castelló-Climent and Rafael Doménech

December 2022

## Converging to Convergence: The Role of Human Capital<sup>\*</sup>

A. Castelló-Climent<sup>a</sup> and R. Doménech<sup>a,b</sup>

a University of Valencia b BBVA Research

December, 2022

#### Abstract

This paper analyses  $\sigma$ -convergence and absolute  $\beta$ -convergence in human capital indicators in a sample of 140 countries from 1970 to 2020. The literature has found some evidence of absolute convergence in per capita income since 2000s. As convergence in human capital preceded convergence in per capita income, human capital may have played a role in this process. We find that the coefficient of  $\beta$ -convergence in per capita income, conditional on the initial level of human capital, is twice as high as the  $\beta$ -coefficient of unconditional convergence. The difference between both coefficients accounts for the role of human capital in explaining absolute convergence in per capita income in recent decades. When we control for institutions, the difference between absolute and conditional convergence is very low, suggesting that human capital has played a greater role in the speed of convergence. The results hold with alternative measures of human capital and development, and are robust to the inclusion of fixed effects and regional dummies to account for omitted variables.

*Keywords*: convergence, GDP per capita, human capital, years of schooling. *JEL Classification*: O11, O15, O4, O47.

\* We thank the comments and suggestions of participants at the SAEe 2022, We are grateful for the financial support of Generalitat Valenciana through the grant PROMETEO /2020/083. A. Castelló would like to thank the Spanish Ministry of Economy and Competitiveness for its financial support through the PGC2018-101161 project. R. Doménech thanks the Ministry of Science and Innovation, PID2020-116242RB-I00, for its financial support. Contact: amparo.castello@uv.es, rafael.domenech@uv.es

#### 1. Introduction

Whether poor countries converge to the income level of rich economies has captured the attention of researchers for many years. Evidence of absolute convergence, in the sense that poor countries unconditionally catch up the income levels of rich economies, has not been found in the literature for many decades (Baumol, 1986; De Long, 1988; Barro, 1991; Pritchett, 1997; Rodrick, 2013). However, new evidence shows signals of unconditional convergence from 2000 onward (Kremer et al., 2021; Patel et al., 2021). In this paper we focus on convergence in human capital, and whether it has played any role in the recent convergence of income per capita observed in the data. Our results show that this is the case. Human capital is not only an important determinant of growth rates but has also helped to accelerate convergence in income levels across countries.

Using data on two measures of human capital, taken from Penn World Table 10 and the latest version of Barro and Lee (2013), we start by analyzing convergence in human capital and whether it preceded convergence in income levels. Our first measure of convergence is  $\sigma$ -convergence. We say there is  $\sigma$ -convergence if there is a tendency for the cross-sectional dispersion to decline over time. Dispersion is measured by the variance across countries of the log of human capital. We find that the variance of human capital started to decline a long time before the fall in the variance of income per capita. In a broad sample of countries, the evidence shows that  $\sigma$ -convergence in human capital started around 1977. Meanwhile, divergence in per capita income between countries was the dominant force until the 1990s. It was only around 2000 when the variance of the log of per capita GDP started to decline. We compute which part of the variance in per capita GDP is determined by the variance of per capita income stabilizes at around 17.0 per cent.

Most of the literature on convergence mainly focuses on  $\beta$ -convergence, that is, a negative correlation between the initial level of a variable and its subsequent growth rates. We estimate the relationship between the human capital index in period t and its growth rate over the next 10 years from 1960 to 2020. We find that  $\beta$ -convergence in human capital has been statistically significant from the 1980s onwards, in line with the evolution of  $\sigma$ -convergence.<sup>2</sup> Unconditional to other factors, countries that started with lower levels of human capital experienced higher growth rates in education. The

<sup>&</sup>lt;sup>2</sup> Barro and Sala-i-Martin (1992) and Young et al. (2008) show that  $\beta$ -convergence is a necessary, but not a sufficient condition for  $\sigma$ -convergence.

coefficient of  $\beta$ -convergence in human capital increases over the years, suggesting a greater convergence process over time. Absolute convergence in human capital has been driven by a slowdown in the growth rates of human capital in the regions at the top (advanced countries and Europe and Central Asian regions), a speeding-up of the growth rates of the regions at the bottom (South-Asia, Sub-Saharan Africa, the Middle East, and North Africa), and a plateauing of the regions in the middle (Latin America and the Caribbean).

As human capital is one of the main determinants of income per capita, the convergence in human capital, starting about 1980, might have played a role in the absolute convergence in per capita income observed in recent decades. As a first step, we corroborate in our dataset the findings in the literature regarding the convergence in income levels. In line with previous studies, we find an absence of absolute convergence in per capita income over a long period of time (Baumol, 1986; Barro, 1991; Pritchett, 1997 and Rodrik, 2013) and a change in the slope from the 2000s (as Kremer et al., 2021). Then, we use the omitted variable formula to analyze the role of human capital in this process. We decompose absolute convergence in two parts: the contribution of conditional convergence ( $\beta$ ) and convergence conditional to human capital ( $\beta^*$ ) is the contribution of human capital. If human capital did not play a role in absolute convergence, we would expect absolute convergence to be equal to conditional convergence ( $\beta = \beta^*$ ).

Our results show that human capital is highly and statistically significant related to the subsequent growth rates in per capita income and is also highly related to the initial level of income. For the period for which there is absolute convergence, 2000s and 2010s, we find that convergence, conditional to the initial level of human capital, is twice as large as absolute convergence. For example, for the decade of 2000s, the coefficient of absolute convergence is -0.575 and that of convergence conditional to human capital is -1.359. Conditioned on human capital, the convergence coefficient increases by 0.784 points.

We show the results are robust to a variety of sensitivity tests. First, Johnson et al. (2013) show that growth rates can vary substantially across datasets. We check whether the results are sensitive to measuring income with the World Development Indicators (WDI). When we use the alternative measure of GDP, the difference between absolute convergence and convergence conditional to human capital increases.

Second, it could be possible that human capital is picking up the role of other determinants of the steady state in the convergence process. One of the most important determinants of the differences in growth rates between countries are institutions. To analyze the role of this determinant, we check the difference between absolute convergence and convergence conditional to institutions, measured by the democracy index, as in Acemoglu et al. (2019), and by the Political Rights Index. Unlike the results with human capital, when we control for institutions, the coefficient of absolute convergence is almost equal to the coefficient of conditional convergence, suggesting that institutions have played a small role in explaining absolute convergence in recent decades.

Third, we test whether the results are driven by the omission of relevant determinants of the growth rates that are constant over time and specific to each country. Acemoglu and Molina (2021) argue that Kremer et al. (2021) findings could be biased due to the omission of country fixed effects that account for unobserved determinants of per capita GDP across countries. The omission of unobserved heterogeneity could bias the convergence coefficients towards zero.<sup>3</sup>

We check the robustness of the results to the estimation of a model that accounts for unobservable heterogeneity. According to the results in the literature, once we control for fixed effects, the convergence coefficient increases and is statistically significant in all decades. The findings also reveal that even controlling for fixed effects, the coefficient of convergence increases when we condition on human capital. The increase in the convergence coefficient is much higher when we condition on human capital than when we condition on institutions.

However, fixed-effect estimators can cause an upward bias in convergence rates in dynamic models with moderate time dimension (Nerlove, 2000). The large increment in the convergence coefficient found with the fixed effect estimator might be due to both the increase in convergence, once we condition for time-invariant determinants of long-run growth, and the upward bias of the fixed effect estimator in a dynamic model. As Barro (2015) shows, there is a trade-off between the omitted variable bias in the OLS model and the upward bias in the fixed-effect estimator.

An alternative approach to control for unobserved heterogeneity, suggested by Temple (1998), is to proxy initial efficiency by controlling for regional dummies. The idea is that the differences in initial efficiency are much greater between regional groups of countries than within regions. When we control for regional dummies the results hold. Again, the convergence coefficient increases when we condition on human capital measures, and

<sup>&</sup>lt;sup>3</sup> The relevance of controlling for country fixed effects in convergence equations was already pointed out by Islam (1995) and Caselli et al. (1996). Islam (2003) and Durlauf et al. (2005) offer an excellent survey of the literature.

the increment is greater than when we condition on institutions.

Most of the literature on convergence has concentrated on convergence in income levels. Our paper adds to the literature by analyzing the role of human capital in this process. Initially, the focus was on absolute convergence in income levels. In a pioneering paper, Baumol (1986) uses data on Madison (1982) for the period 1870-1979 and finds absolute convergence in output per labor among 16 industrialized nations but finds no evidence of convergence in GDP per capita in a broader sample of 72 countries from 1950 to 1980. De Long (1988) showed that Baumol's finding on absolute convergence for industrialized countries is driven by selection bias and measurement error in the independent variable.<sup>4</sup> Once both considerations are taken into account, absolute convergence is absent even in the set of rich economies. In a broader sample of 98 countries, Barro (1991) confirms this finding and shows that per capita growth rates are not correlated with the starting levels of GDP per capita during the period 1960-1985. Pritchett (1997) found that convergence between the poorest and the richest economies did not occur between 1870 and 1990, suggesting that there has been divergence for long-time horizons.<sup>5</sup>

Instead of absolute convergence, for many years the literature focused on conditional convergence, in the sense that poor countries converge to their steady states after conditioning on institutions, policies or other country-specific factors. Barro and Salai-Martin (1992) and Mankiw, Romer and Weil (1992) were among the first to show that conditioning on variables that proxy for determinants of the steady state renders a negative and statistical significant coefficient of initial income on growth in a broad sample of countries. Absolute or unconditional convergence has been found only in countries or regions that share similar long-run characteristics or steady-state determinants (see Barro and Sala-i-Martin, 1992), but not in a broad sample of countries (Barro, 1991, Pritchett, 1997, or Rodrik, 2013).

The lack of evidence has rendered a rejection of the absolute convergence hypothesis during several decades.<sup>6</sup> Nevertheless, the new evidence by Kremer et al. (2021) and Patel et al. (2021) shows some signals of unconditional convergence in per capita income in recent decades. In this paper we show that convergence in human capital has played an important role in the recent convergence in income levels observed in the data.

<sup>&</sup>lt;sup>4</sup> Baumol's sample is criticized as being an ex post sample of countries that have converged. Countries that were relatively rich in 1870 but did not converge, such as Chile, Argentina, Spain, and Portugal, were not included in the sample of the 16 industrialized nations.

<sup>&</sup>lt;sup>5</sup> More recently, Rodrik (2013) shows strong unconditional convergence in labor productivity in manufacturing, but an absence of absolute convergence in per capita GDP in a broad sample of countries.

<sup>&</sup>lt;sup>6</sup> Johnson and Papageorgiou (2020) present and excellent survey of the literature.

Evidence on the relationship between human capital convergence and income convergence has been scarce in the literature. Using data for 84 countries for the period 1970-1990, Sab and Smith (2002) find convergence in human capital, measured with indicators that include both education and health. The paper, however, does not analyze whether convergence in human capital relates to convergence in income levels.

In a pioneering paper, O'Neill (1995) found that, between 1967 and 1985, human capital convergence explained income convergence among developed countries. However, despite the convergence in educational levels, the increase in the skill premium led to divergence in income levels in the least developed countries and in the world as a whole. Our paper focuses on a more recent period and uses a measure of human capital that takes into account not only the years of schooling but also the return of each level of education. For the years 2000-2020, for which absolute convergence has been found, we show that convergence in human capital has indeed helped to explain convergence in income levels in a broad sample of countries with different levels of development.

The structure of the paper is as follows. Section 2 analyzes  $\sigma$ -convergence in human capital and in per capita income and checks to what extent the variance in human capital explains the variance in per capita income. Section 3 presents the results of  $\beta$ -convergence in human capital, and analyzes the role of human capital in the process of absolute convergence in per capita income found in the recent decades. Section 4 analyzes the robustness of the results. Finally, Section 5 concludes.

#### 2. $\sigma$ -convergence

In this paper, we rely on two basic data sources. Most of the data, as GDP and population, is taken from Penn World Table 10 (see Feenstra, Inklaar and Timmer, 2015). Since PWT version 8, this data set also includes an estimate measure of human capital ( $h_{PWT}$ ) for a large sample of countries, which is based on the approach proposed by Hall and Jones (1999) and Caselli (2005).

$$h_{PWT} = e^{\phi(s)} \tag{1}$$

where *s* is the average years of education of the population 15-64 years old, and  $\phi(s)$  is a linear function that aggregates the years of schooling for primary, secondary, and tertiary education. In particular,  $\phi(s) = 0.134s$  if  $s \le 4$ ,  $\phi(s) = 0.134s + 0.101(s - 4)$  if  $4 < s \le 8$ , and  $\phi(s) = 0.134s + 0.101s + 0.068(s - 8)$  if s > 8 The coefficients  $\phi(s)$  are taken from Psacharopoulos (1994) and correspond to the rates of return of an additional year

of education for the three levels of education considered.

PWT combines two data sets for years of schooling. For 95 countries, years of schooling are based on Barro and Lee (2013) data (BL henceforth). For 55 countries, data are based on Cohen and Leker (2014) estimates (CL henceforth). As shown by De la Fuente and Doménech (2006) and Cohen and Soto (2007), the Barro and Lee (2013) estimates are rather noisy and are subject to some breaks over time and measurement errors. In PWT the choice of the data source depends on which series is closer to De la Fuente and Domenech (2006) or the UNESCO data, if a series for a country shows a decline in the average year of schooling, or if data start in 1950 versus 1960. The BL data are available every five years from 1950 to 2015, whereas the CL data are available every 10 years, from 1960 to 2010, but include projections to 2020. PWT interpolates linearly between observations, and extrapolates BL data to 2019 using the growth rates of CL, if they are available, or, alternatively, the trend growth of BL from 2005 to 2015.

For comparisons, we follow the same approach as in PWT to estimate  $h_{BL}$  using only the original Barro and Lee (2013) data. The correlation between  $h_{PWT}$  and  $h_{BL}$  in log. levels is 0.96, but falls to 0.65 in rates of growth.

As we need to combine human capital with other variables in PWT, the sample of countries with available data changes over time. Thus, we have 55 countries with complete information from 1950 to 1969, 99 from 1960 to 1969, 124 from 1970 to 1989, and 140 from 1990 to 2019.

Figure 1 shows the variance of the log of  $h_{BL}$  for the different samples of countries that start in different decades. As we can observe, the evolution over time is very similar for all samples, with the exception of the sample of 55 countries starting in 1950, where the convergence process has been more intense from late 1970s onwards. Given these similarities, we can extrapolate backwards the variance of  $\ln h_{BL}$  for the whole sample of countries from 1950 to 2015, using the rate of change of the variance of the smaller samples of countries. Figure 1 shows that this variance has been decreasing from 1977 onward.

Figure 2 shows the variance of the log of  $h_{PWT}$ , for the same samples of countries. Again, the evolution over time is highly analogous for all samples, except for the sample of 55 countries. The main differences with respect to  $\ln h_{BL}$  are that the level of the variance of  $\ln h_{PWT}$  is slighter greater and the convergence process is not as intense as with the BL data.

<sup>&</sup>lt;sup>6</sup> We rely on years of schooling since this variable is available for more countries and years than quality indicators of human capital, as discussed by De la Fuente and Domenech (2022).



**Figure 1:** *Variance of*  $h_{BL}$  *in logs.* 



**Figure 2:** Variance of  $h_{PWT}$  in logs.

As human capital is one of the main determinants of income per capita, the convergence process shown in Figures 1 and 2 suggests the convenience of testing its effects on the convergence of GDP per capita (y). In this section we analyse the success of the variance of  $\ln h_{PWT}$  and  $h_{BL}$  in explaining the variance of  $(\ln y)$  across countries, whereas in the next section we test the relevance of both indicators of human capital in terms of the  $\beta$ -convergence hypothesis.

For simplicity, we assume that GDP per capita for country i and time t can be expressed in terms of the following Cobb-Douglas function

$$y_{it} = A_{it} k_{it}^{\alpha} h_{it}^{1-\alpha} \tag{2}$$

where *A* is total factor productivity (TFP) and *k* the capital to labor ratio. As in Klenow and Rodriguez-Clare (1997) and Hall and Jones (1999), we can rearrange terms and write the production function in terms of the capital-to-output ratio:

$$y_{it} = h_{it} A_{it}^{\frac{1}{1-\alpha}} \left( k_{it} / y_{it} \right)^{\frac{\alpha}{1-\alpha}}$$
(3)

Defining the term  $A_{it}^{\frac{1}{1-\alpha}} (k_{it}/y_{it})^{\frac{\alpha}{1-\alpha}}$  as  $y_{AKY}$ , then we have:

$$\operatorname{var}(\ln y_t) = \operatorname{var}(\ln h_t) + \operatorname{var}(\ln y_{AKYt}) + 2\operatorname{cov}(\ln y_{AKYt}, \ln h_t)$$
(4)

Taking into account that  $\sigma$ -convergence refers to the evolution over time of the standard deviation of the log of GDP per capita across countries (see Barro and Sala-i Martin 1992), equation (4) shows the contribution of the convergence of human capital to convergence of income, measured in terms of its variance instead of the standard deviation.

As in the case of  $\ln h_{PWT}$  and  $\ln h_{BL}$ , we begin analysing the sensitivity of the variance of the log of GDP per capita to changes in the sample of countries.<sup>7</sup> As we can see in Figure 3, the composition of countries in the four different samples affects the level of the variances, although not so much their time profiles. As in the case of human capital, we extrapolate backward the variance of  $\ln y$  for the entire sample of countries from 1950 to 2015, using the variance of the smaller sample of countries.

As noted by Kremer et al. (2021), Figure 3 shows that divergence between countries was the dominant force until the 1990s. In their research, they use an unbalanced panel with all countries with data on GDP per capita available. This approach intensifies divergence until 1970. In our case, the backward extrapolation of the variance with the larger

<sup>&</sup>lt;sup>7</sup> GDP per capita is measured using PWT variable RGDPNA.



Figure 3: Variance of GDP per capita in logs.

sample of 140 countries, using the rate of change of smaller samples, makes divergence less pronounced. The Great Recession marked the start of a process of convergence for eight years until 2016 that is common for the four different samples of countries, partially reverted in the last three years of the sample.

Now we calculate which part of the variance of  $\ln y$  is determined by the variance of human capital. To this end, we use  $h_{PWT}$  as a measure of human capital and compute two ratios. The first is a strict measure of its contribution, given by the ratio of the variance of human capital to the variance of GDP per capita:

$$Contribution_1 = \frac{\operatorname{var}(\ln h_{PWT})}{\operatorname{var}(\ln y)},\tag{5}$$

whereas the second measure also includes the covariance between human capital and the rest of the determinants of GDP per capita in equation (3), that is, the stock of physical capital and TFP.

$$Contribution_2 = \frac{\operatorname{var}(\ln h_{PWT}) + \operatorname{cov}(\ln h_{PWT}; \ln y_{AKY})}{\operatorname{var}(\ln y)}$$
(6)

In Figure 4 we represent both measures of the contribution of human capital to the variance of the log of GDP per capita. The breaks in 1960, 1970 and 1990 are due to the changes in the samples from 53 to 99, 124 and 140 countries, respectively. We can see that



Figure 4: Contribution of human capital to the variance of the log of income per capita

in the case of the sample of 53 countries in the 1950s, human capital contributes around 12 percent to the variance of per capita income.

As expected, the covariance between human capital and  $\ln y_{AKY}$  is positive, thus helping to increase the contribution of human capital to the variance of per capita income to around 28.5 per cent in the 1950s. As we increase the number of countries from 53 to 99 in the 1960s and to 124 in the 1970s, we observe a significant reduction in the contribution of human capital. Therefore, the composition of the sample of countries is very relevant in order to determine the contribution of different production factors to GDP per capita. From 1970 onward, the contribution of human capital stabilizes at around 17.0 per cent, according to the second measure that includes the covariance component.<sup>8</sup>

#### 3. $\beta$ -convergence

As is well known, there is  $\beta$ -convergence across countries when the growth rate of a variable is inversely related to its initial value. As preliminary evidence, Figure 5 shows the relationship between the PWT human capital index in the period *t* and its growth

<sup>&</sup>lt;sup>8</sup> As a robustness test, we have computed the contribution of human capital to the variance of the log of GDP per capita using PWT variable RGDPo instead of RGDPNA. The contribution increases slightly in the 1950s to 29.3 per cent, and stabilizes at around 18.2 per cent from 1970 onwards.

rate over the next 10 years t, t + 10. There is  $\beta$ -convergence when the slope of this relationship is negative and statistically significant. With the exception of the 1960s and the 1970s, the figure shows a clear negative slope in all decades. Countries that start with lower levels of education in the base year tend to growth faster over the subsequent decade, suggesting that there has been human capital convergence over a long time horizon. Although  $\beta$ -convergence is not a sufficient condition for  $\sigma$ -convergence (see Barro and Sala-i-Martin, 1992, 227-8), the evidence presented in Figure 5 is consistent with the convergence process in Figure 2 from 1980 onward.

Table 1 analyzes this evidence by testing the statistical significance of the slopes in Figure 5. The dependent variable is the average growth rate of human capital over each decade in the sample. The explanatory variables include human capital at the beginning of each decade and dummy variables that control for time effects. We display the results for the human capital index in PWT,  $\ln h_{PWT}$ , and the human capital index computed with the average years of schooling from Barro and Lee (2013),  $\ln h_{BL}$ .

Columns (1) and (3) show the result for the whole period.<sup>9</sup> The pooled OLS regression model shows a strong correlation between the initial level of human capital and its growth rate. The coefficient of the initial human capital is negative and statistically significant at the 1 per cent level for both measures of human capital. In columns (2) and (4) we include interaction terms between the level of human capital at the beginning of each period and a year dummy ( $y_t$ ) to analyze how convergence has changed over time. This specification provides the same results as estimating cross-section regressions individually for each decade. Convergence in human capital starts around 1980 for the PWT variable and in 1970 for Barro and Lee. The coefficients of the interaction terms are statistically significant at the 1 per cent level from 1980 onward for  $\ln h_{PWT}$  (column (2)) and from 1970 onward for  $\ln h_{BL}$  (column (4)). The evidence also shows increasing coefficients over the years, suggesting a greater convergence process over time.

The convergence in human capital could have occurred because countries that started with high levels of human capital have seen their growth rates slow, because countries with low initial levels have seen their growth rates increase, or both. Table 2 shows the average rate of human capital growth in each decade for seven geographical regions with different initial levels of human capital. <sup>10</sup>

<sup>&</sup>lt;sup>9</sup> Barro and Lee's (2013) data is available up to 2015. For the decade 2010-2020, the average growth rate of  $h_{BL}$  is calculated from 2010 to 2015.

 $<sup>^{10}</sup>$  In Table 2 we use the human capital index computed with Barro and Lee's (2013) data because it is a balanced panel with 140 countries from 1960 to 2015.



**Figure 5:**  $\beta$ -convergence in human capital (PWT)

· ·	0		-	
Dep. variał	ole: decadal	average hum	an capital gro	wth rate
	(1)	(2)	(3)	(4)
	$h_{P}$	WT	$h_1$	3L
ln <i>h</i>	-0.390***		-0.588***	
	(0.062)		(0.081)	
$\ln h \times y_{1960}$		0.474***		0.124
		(0.123)		(0.191)
$\ln h \times y_{1970}$		-0.105		-0.356**
		(0.166)		(0.179)
$\ln h \times y_{1980}$		-0.590***		-0.796***
		(0.165)		(0.187)
$\ln h \times y_{1990}$		-0.536***		-0.532***
		(0.133)		(0.164)
$\ln h \times y_{2000}$		-0.628***		-1.090***
		(0.126)		(0.146)
$\ln h \times y_{2010}$		-0.780***		-2.015***
		(0.202)		(0.416)
Constant	0.689***	0.489***	0.845***	0.614***
	(0.072)	(0.067)	(0.054)	(0.040)
Obs.	821	821	980	980
$R^2$	0.122	0.155	0.137	0.208
Year FE	yes	yes	yes	yes

**Table 1**:  $\beta$ -convergence in human capital, 1960-2020

Robust standard errors in parentheses

\*\*\* p<0.01, \*\* p<0.05, \* p<0.1

In 1960, the highest level of human capital was concentrated in advanced economies, followed by Europe and Central Asia. Latin America and the Caribbean countries came third, followed by the countries in East Asia and the Pacific region. At the bottom were countries in Middle East and North Africa, South Asia, and Sub-Saharan Africa, respectively. During this decade there was a slowdown in the growth rates of human capital in advanced economies and a speeding-up in the growth rates in East Asia and the Pacific region. However, there was no convergence at the bottom. South Asia and Sub-Saharan African countries had the lowest initial levels of human capital and also show the lowest growth rates.

From 1970 onward, advanced economies, which were ranked as the region with the highest initial levels of human capital, show the lowest growth rates across almost all

		Table 2	2: Decadal A	werage H	uman Ca	pital Grov	vth Rate b	y Geograph	ical Region						
					Growt	h Rate					×	anking			
	Countries	HC <sub>1960</sub>	1960s	1970s	1980s	1990s	2000s	2010s	HC1960	1960s	1970s	1980s	1990s	2000s	2010s
Whole sample	140	1.642	1.031	1.222	1.072	0.875	0.832	0.779							
Advanced Economies	24	2.226	0.923	1.021	0.681	0.651	0.635	0.258	1	5	9	7	7	9	4
East Asia and the Pacific	17	1.509	1.387	1.423	1.008	0.967	1.129	0.621	4	1	2	ß	ю	2	5
Europe and Central Asia	20	2.078	1.191	1.330	0.809	0.765	0.450	0.305	2	2	б	9	9	7	9
Latin America and the Caribbean	24	1.629	1.036	1.179	1.139	0.928	0.763	0.670	Ю	4	4	4	4	ß	4
Middle East and North Africa	17	1.370	1.179	1.544	1.612	1.184	0.951	1.341	9	ю	1	1	2	4	2
South Asia	9	1.309	0.910	0.821	1.176	1.199	1.232	1.463	9	9	4	ю	1	1	1
Sub-Saharan Africa	32	1.217	0.764	1.134	1.206	0.798	0.975	1.203	7	7	5	2	5	3	3

	ē
- 1	2
1	a
	2
	5
	8
	5
	ы
	2
1	٠ų
1	-
	2
	4
	ğ
	~
	9
	ぢ
	R
	Ч
- (	5
	_
	E
	7
	a
- (	Ű
	2
	a
	R
	Ξ
,	÷
	Я
	š
	H
	5
	∢
	_
	10
	S
	Ŭ
	ě
1	
	÷
	a
	ź

decades, while regions at the bottom start speeding up. Sub-Saharan African countries jump from the 7th position in the 1960s to the 3rd position in the 2000s, and countries in the Middle East and North Africa show one of the highest growth rates over the years.

Countries in South Asia deliver one of the most striking performances. Whereas this region had the lowest growth rates in the 1970s, it jumps up to third position in the 1980s, and from the 1990s onward, South Asia is the region with the highest growth rates during three decades. Meanwhile, there is hardly any movement in the middle of the distribution. Countries in the Latin America and Caribbean region hold a fourth position in terms of growth rates in almost all decades.

Overall, this evidence indicates that  $\beta$ -convergence in human capital started mainly around 1980. Convergence in human capital has been driven by a slowdown in the growth rates of regions at the top, a speeding-up of the growth rates of the regions at the bottom, and a plateauing of regions in the middle.

#### 3..1 Human capital and $\beta$ -convergence in GDP per capita

Has convergence in human capital played a role in the process of convergence in per capita GDP observed in recent decades? To answer this question, we estimate three equations. Firstly, we analyze the relationship between initial income and growth using the following empirical framework:

$$\ln(y_{i,t+\triangle t}) - \ln(y_{i,t}) = \alpha + \beta_t \ln(y_{i,t}) + \mu_t + \epsilon_{i,t}$$
(7)

where *y* is the real GDP per capita in the country *i* in year t,<sup>11</sup>  $\mu_t$  is a year fixed effect that controls for shocks that are common to all countries, and  $\epsilon_{i,t}$  is an error term that varies across countries and years. We assume that  $\epsilon_{i,t}$  is uncorrelated with the explanatory variables. The convergence coefficient  $\beta$  is allowed to vary over time. In this framework, the estimate of  $\beta$  will be a measure of unconditional or absolute convergence. There is unconditional convergence in GDP per capita if  $\beta < 0$ .

Then, we analyze whether the relationship between initial income and growth changes conditioning on initial human capital. To do so, we include a measure of human capital in the set of controls:

$$\ln(y_{i,t+\Delta t}) - \ln(y_{i,t}) = \alpha + \beta_t^* \ln(y_{i,t}) + \lambda_t \ln h_{i,t} + \mu_t + \varepsilon_{i,t}$$
(8)

<sup>11</sup> The variable is the real GDP per capita at constant 2017 national prices (in million 2017 USD) from the PWT10. In the following section we analyze the robustness of the results with alternative measures of GDP.

where  $lnh_{i,t}$  is the measure of human capital in the country *i* at the beginning of the period. The effect of human capital on GDP per capita growth,  $\lambda_t$ , is allowed to vary over time. In this expression, the coefficient  $\beta^*$  captures convergence conditional to controlling for the initial level of human capital. There will be conditional convergence if  $\beta^* < 0$ .

If we assume that human capital is a determinant of GDP per capita growth and we omit this variable in the set of controls, we can decompose the difference between absolute and conditional convergence through the omitted variable bias formula. An omitted variable bias exists if human capital is a determinant of economic growth and if it is correlated with ln *y*. If human capital and GDP per capita are correlated as follows:

$$\ln h_{i,t} = \phi + \delta_t \ln(y_{i,t}) + \mu_t + v_{it} \tag{9}$$

then, we can substitute (9) into (8) and we can decompose absolute convergence ( $\beta_t$ ) as the contribution of conditional convergence ( $\beta_t^*$ ) and the contribution of human capital, measured by the product of the effect of human capital on growth ( $\lambda_t$ ) and the coefficient of GDP per capita ( $\delta_t$ ) in equation (9):

$$\beta_t = \beta_t^* + \lambda_t \times \delta_t \tag{10}$$

If human capital did not play any role in absolute convergence, we would expect  $\lambda_t \times \delta_t = 0$  and  $\beta_t = \beta_t^*$ . If human capital affects convergence and  $\lambda > 0$  and  $\delta > 0$ , absolute convergence will occur if  $\beta^* < 0$  and  $\beta^* > (\lambda \times \delta)$ .

Table 3 displays the results of estimating equations (7)-(9). Equation (7) assumes that initial conditions, apart from initial income, do not matter for growth. In line with the findings in the literature, the results in column (1) corroborate the absence of absolute convergence from 1960 to 2000 (Barro, 1991; Pritchett, 1997; or Rodrik, 2013), and a change in the slope thereafter (Kremer et al., 2021). For the period 2000-2010, the estimated coefficient of  $\beta$  is negative and statistically significant at the one percent level, suggesting that a process of absolute convergence might have started from the 2000s onward. Independently of initial conditions, countries that started with lower GDP per capita in the year 2000, experienced more rapid growth rates over the subsequent decade. Results also display a negative estimate of  $\beta$  for the period 2010-2019, though the coefficient is not statistically significant at the standard levels.

We check whether the results are driven by atypical observations. In column (2) we include dummies that control for outliers.<sup>12</sup> Whereas controlling for outliers hardly

<sup>&</sup>lt;sup>12</sup> We include dummies for those countries and years for which the residuals are three times higher than their

		De	ecada	al Average	Human (	Capital Gro	ow.	th Rate		
	(1)	(2)		(3)	(4)	(5)		(6)	(7)	(8)
					$h_{PWT}$				$h_{BL}$	
	β	β		$\beta^*$	λ	δ		$\beta^*$	λ	δ
1960s	0.336 <sup>c</sup>	0.336 <sup>c</sup>		-0.653 <sup>a</sup>	4.283 <sup>a</sup>	0.231 <sup>a</sup>		-0.580 <sup>c</sup>	4.392 <sup>a</sup>	0.208 <sup>a</sup>
	(0.183)	(0.183)		(0.346)	(1.212)	(0.021)		(0.312)	(1.210)	(0.019)
1970s	0.030	0.236		-0.215	$2.717^{a}$	$0.171^{a}$		-0.281	3.427 <sup>a</sup>	$0.155^{a}$
	(0.264)	(0.189)		(0.252)	(0.954)	(0.026)		(0.242)	(1.009)	(0.022)
1980s	-0.203	-0.252		-1.232 <sup>a</sup>	$5.841^{a}$	0.173 <sup>a</sup>		$-1.201^{a}$	6.489 <sup>a</sup>	$0.151^{a}$
	(0.248)	(0.245)		(0.293)	(0.953)	(0.021)		(0.283)	(1.089)	(0.019)
1990s	0.114	0.014		-0.012	0.134	0.195 <sup>a</sup>		0.063	-0.293	$0.171^{a}$
	(0.206)	(0.175)		(0.268)	(0.988)	(0.015)		(0.251)	(1.159)	(0.014)
2000s	$-0.575^{a}$	-0.575 <sup>a</sup>		$-1.359^{a}$	$4.203^{a}$	$0.174^{a}$		$-1.190^{a}$	4.013 <sup>a</sup>	$0.153^{a}$
	(0.164)	(0.166)		(0.241)	(0.866)	(0.015)		(0.258)	(1.039)	(0.014)
2010s	-0.173	$-0.440^{a}$		$-0.904^{a}$	$2.454^{a}$	0.183 <sup>a</sup>		$-0.965^{a}$	3.275 <sup>a</sup>	$0.155^{a}$
	(0.238)	(0.142)		(0.195)	(0.845)	(0.015)		(0.187)	(0.974)	(0.013)
Constant	2.366 <sup>a</sup>	2.366 <sup>a</sup>		2.366 <sup>a</sup>	2.366 <sup>a</sup>	0.513 <sup>a</sup>		2.366 <sup>a</sup>	2.366 <sup>a</sup>	$0.504^{a}$
	(0.273)	(0.275)		(0.277)	(0.277)	(0.044)		(0.277)	(0.277)	(0.040)
R <sup>2</sup>	0.101	0.313		0.403	0.403	0.621		0.397	0.397	0.643
Obs	821	821		821	821	821		821	821	821
Year FE	YES	YES		YES	YES	YES		YES	YES	YES
Dummy outliers	NO	YES		YES	YES	YES		YES	YES	YES

**Table 3**: Conditional and unconditional  $\beta$ -convergence

a, b and c denote statistically significant coefficients at 1, 5 and 10 per cent probabilities, respectively.

changes the results in most of the decades, the coefficient of  $\beta$  convergence increases and becomes statistically significant for the years 2010-2019.<sup>13</sup> Thus, in line with the literature, the results suggest an absence of absolute convergence before the year 2000, and evidence of convergence afterwards.

How do the results change when we condition for the initial level of human capital? The results in column (3) show that the coefficient of  $\beta^*$ —convergence is negative in all periods, although it is not statistically significant in all decades. Column (4) shows that the initial level of human capital is an important determinant of growth rates. With the exception of the 1990s, human capital has a positive effect on subsequent growth rates. In all periods  $\lambda_t$  is positive and statistically significant at the 1 percent level. Human capital is also highly related to the initial level of income. The estimates of  $\delta$  are positive and

standard deviation. The dummies include Botswana (1970), Cambodia (1970), Kuwait (1970), Liberia 1980), Moldova (1990), Tajikistan (1990), Ukraine (1990), D.R. of the Congo (1990), Sierra Leone (1990), Venezuela (2010) and Yemen (2010).

<sup>&</sup>lt;sup>13</sup> Venezuela is a clear outlier for the years 2010-2019, with the lowest GDP per capita in 2010, and a growth rate of -11.16 percent during this period.

		Iddie 4. L	ccompos		weigenee		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
			HC PV	NT		HC E	3L
	β	$\beta^*$	$(\lambda  imes \delta)$	$\beta^* + (\lambda  imes \delta)$	$\beta^*$	$(\lambda  imes \delta)$	$eta^* + (\lambda  imes \delta)$
2000s	-0.575	-1.359	0.731	-0.628	-1.190	0.614	-0.576
2010s	-0.440	-0.904	0.449	-0.455	-0.965	0.508	-0.457

**Table 4**: Decomposition of  $\beta$ -convergence

statistically significant at 1 per cent level in almost all periods (column (5)). The results are similar to the measure of human capital computed with Barro and Lee's (2013) data (column (6)-(8)). It is worth noting that  $\lambda$  is above 2 and statistically significant in all decades, except in the 1990s. Meanwhile,  $\delta$  is around 0.18 and statistically significant in all decades.

Table 4 summarizes the decomposition of absolute  $\beta$ -convergence into conditional convergence and the role of human capital, in the two decades in which we observe absolute convergence, that is, from 2000 onward. Columns (1) and (2) show that conditional convergence is twice as large as absolute convergence. The difference between absolute and conditional convergence ( $\beta - \beta^*$ ) is the role of human capital ( $\lambda \times \delta$ ). For the period 2000-2010, the coefficient of absolute convergence ( $\beta = -0.575$ ) more than doubles once we condition for human capital ( $\beta^* = -1.359$ ), suggesting that human capital accelerates the process of income convergence across countries during that period. In quantitative terms, the role of human capital in the convergence process is the difference between both coefficients ( $\beta - \beta^* = 0.784$ ). Once human capital is controlled for, the convergence coefficient also doubles for the period 2010-2019. Results hold when we measure human capital with Barro and Lee's (2013) measures. In what follows, we check the robustness of these findings.

#### 4. Robustness

#### 4..1 Alternative measures of GDP

In this paper we measure income levels with data from Penn World Tables, which is the most widely used source of comparable data on GDP for a broad number of countries and periods. The variable we use is the Real GDP per capita at constant 2017 national prices (in million 2017 USD) from PWT10. Johnson et al. (2013) show that the estimates of GDP might vary substantially across versions of the Penn World Tables. When they analyze the

		Alternativ	e measure c	of GDP (W	orld Devel	opment Indi	cators)	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
				HC PWT			HC BL	
	β	β	$\beta^*$	λ	δ	$\beta^*$	λ	δ
1960s	0.551 <sup>a</sup>	$0.551^{a}$	0.123	2.209	0.194 <sup><i>a</i></sup>	0.069	2.840	0.170 <sup>a</sup>
	(0.146)	(0.150)	(0.386)	(1.872)	(0.017)	(0.333)	(1.856)	(0.015)
1970s	0.149	0.216	-0.320	$2.864^{c}$	0.187 <sup>a</sup>	-0.380	$3.566^{b}$	0.167 <sup>a</sup>
	(0.144)	(0.132)	(0.318)	(1.486)	(0.015)	(0.284)	(1.454)	(0.014)
1980s	0.086	0.165	$-0.887^{a}$	5.719 <sup>a</sup>	$0.185^{a}$	$-0.851^{a}$	6.213 <sup>a</sup>	0.161 <sup><i>a</i></sup>
	(0.200)	(0.191)	(0.327)	(1.274)	(0.013)	(0.310)	(1.436)	(0.012)
1990s	$0.281^{c}$	0.213	0.016	1.185	0.164 <sup><i>a</i></sup>	0.033	1.252	$0.141^{a}$
	(0.153)	(0.130)	(0.270)	(1.296)	(0.011)	(0.246)	(1.393)	(0.011)
2000s	$-0.464^{a}$	$-0.424^{a}$	-1.219 <sup>a</sup>	$5.124^{a}$	0.149 <sup>a</sup>	$-1.188^{a}$	5.590 <sup>a</sup>	0.131 <sup>a</sup>
	(0.133)	(0.120)	(0.165)	(0.742)	(0.010)	(0.158)	(0.835)	(0.010)
2010s	$-0.372^{a}$	$-0.308^{a}$	$-0.942^{a}$	3.792 <sup>a</sup>	0.150 <sup>a</sup>	-0.927 <sup>a</sup>	4.373 <sup>a</sup>	0.127 <sup>a</sup>
	(0.119)	(0.102)	(0.198)	(0.952)	(0.011)	(0.184)	(1.066)	(0.010)
Constant	-1.283	-1.283	1.120	1.120	$-1.087^{a}$	1.232	1.232	$-0.885^{a}$
	(1.160)	(1.169)	(2.226)	(2.226)	(0.132)	(1.857)	(1.857)	(0.118)
	0.138	0.313	0.381	0.381	0.719	0.374	0.374	0.728
Obs	664	664	664	664	664	664	664	664
Year FE	YES	YES	YES	YES	YES	YES	YES	YES
Dummy outliers	NO	YES	YES	YES	YES	YES	YES	YES

**Table 5**: Unconditional and conditional  $\beta$ -convergence using WDI GDP

robustness of the results of relevant papers in the growth literature across datasets, they find that studies that use long-run growth rates (30-year averages) are robust across PWT versions, whereas results based on annual data are not. We check whether our findings, based on growth rates over a 10-year span, hold with different measures of GDP.

Table 5 displays the results of estimating equations (7)-(9) with the measure of GDP per capita from the World Development Indicators (WDI). Results hold. In line with previous findings, the measure of GDP from the WDI also shows absolute convergence in GDP per capita from the 2000s onward. When we control for human capital, the co-efficient of conditional convergence triples during the 2000s and 2010s, suggesting that human capital played a relevant role in the convergence process.

<sup>13</sup> Data are in constant 2015 prices, expressed in U.S. dollars.

#### 4..2 Alternative controls

Conditional on determinants of the steady state, several papers have found evidence of conditional convergence (Barro and Sala-i-Martin, 1992; Mankiw, Romer and Weil, 1992). Although human capital is one of the main determinants of long-term growth rates, it has been argued that institutions are the fundamental cause of economic growth and the main reason for differences in economic development between countries (Acemoglu et al., 2005).

There has been much discussion about whether it is necessary first to increase the human capital of a society in order to improve the quality of its institutions , or whether the causality goes in the opposite direction, that is, whether the influence of institutional quality on well-being is due to the fact that it increases the incentive to devote resources and personal effort to the accumulation of knowledge. Glaeser et al. (2004), based on the work of Lipset (1960), consider that the causality goes from human capital to institutional quality. Conversely, Acemoglu et al. (2014) argue that the causality moves in the opposite direction, giving as an example the case of the first European settlers, for whom their low educational level did not suppose a significant limitation in the subsequent development of more advanced institutions in the colonies. In general, both determinants interact in a complementary and dynamic way and are able to generate a virtuous circle. Good institutions are essential to harness all the human capital that a society accumulates. In the same way, it is much easier for institutions to function better if society has the necessary human capital to do so.

Since in our analysis of conditional convergence we have only controlled for human capital, it could be possible that the role of human capital on the process of convergence is picking up the effect of institutions. To analyze this possibility, we compare the difference between absolute and conditional convergence ( $\beta - \beta^*$ ) when we condition on human capital and when we alternatively control for institutions.

We measure institutions with two standard variables in the empirical growth literature. The first one is the dichotomous measure of democracy computed by Acemoglu et al. (2020). This variable is available annually from 1960 to 2010. The second one is the Freedom House Political Rights Index, which is available from 1972 onwards. The variable ranges from 1 to 7, indicating more freedom the closer the value is to 1. We nor-

<sup>&</sup>lt;sup>13</sup> Results are similar if instead of the political rights index we measure institutions with the civil liberties index.

<sup>&</sup>lt;sup>13</sup> The ratings are determined by a checklist of 25 questions. The political rights questions are grouped into three subcategories: electoral process, political pluralism and participation, and the functioning of govern-

			Сс	nditioning	g or	n Institutio	ons		
	(1)	(2)	(3)	(4)		(5)	(6)	(7)	(8)
		Democra	cyANRR				Politica	l Rights	
	β	$\beta^*$	λ	δ		β	$eta^*$	λ	δ
1960s	0.332	0.165	0.651	0.256 <sup>a</sup>					
	(0.234)	(0.256)	(0.551)	(0.052)					
1970s	$0.393^{b}$	0.556 <sup>a</sup>	-0.773	$0.188^{a}$					
	(0.162)	(0.209)	(0.549)	(0.041)					
1980s	-0.200	$-0.516^{b}$	$2.431^{a}$	$0.135^{a}$		-0.227	$-0.607^{b}$	$3.152^{a}$	$0.127^{a}$
	(0.247)	(0.246)	(0.586)	(0.037)		(0.251)	(0.271)	(0.815)	(0.028)
1990s	0.017	-0.146	$1.141^{a}$	$0.144^{a}$		0.078	-0.126	$1.410^{b}$	$0.145^{a}$
	(0.178)	(0.200)	(0.419)	(0.030)		(0.175)	(0.216)	(0.637)	(0.026)
2000s	-0.629 <sup>a</sup>	-0.686 <sup>a</sup>	0.599	0.095 <sup>a</sup>		$-0.570^{a}$	-0.661 <sup>a</sup>	0.764	$0.120^{a}$
	(0.160)	(0.170)	(0.457)	(0.034)		(0.151)	(0.181)	(0.612)	(0.025)
2010s	$-0.436^{a}$	$-0.497^{a}$	0.799 <sup>c</sup>	$0.086^{b}$		-0.416 <sup>a</sup>	-0.559 <sup>a</sup>	1.161 <sup>c</sup>	0.123 <sup>a</sup>
	(0.146)	(0.144)	(0.424)	(0.034)		(0.145)	(0.157)	(0.630)	(0.022)
Constant	0.358	1.456	1.456	-1.686		2.663	$4.438^{b}$	$4.438^{b}$	-0.638 <sup>a</sup>
	(2.071)	(2.126)	(2.126)	(0.455)		(2.159)	(2.108)	(2.108)	(0.244)
R <sup>2</sup>	0.341	0.392	0.392	0.193		0.348	0.403	0.403	0.223
Obs	724	724	724	724		529	529	529	529
Year FE	YES	YES	YES	YES		YES	YES	YES	YES
Dummy outliers	YES	YES	YES	YES		YES	YES	YES	YES

**Table 6**: Unconditional and conditional  $\beta$ -convergence: The role of institutions

malize the variable to range from 0 to 1, where the greater the value the more democratic a set of institutions is.

Table 6 shows the results of estimating equations (7)-(9) with two measures of institutions for the decades with available data. As the number of observations has changed, we first estimate equation (7) in the reduced samples with available data for institutions. In line with previous findings, (columns (1) and (5)) show a lack of absolute convergence up to the 1990s, and evidence of unconditional convergence from 2000 onward.

The role of institutions is analyzed in the remaining columns. As expected, institutions are an important determinant of the growth rates, revealing that more democratic institutions are associated with higher per capita GDP growth (columns (3) and (7)). However, the effect of institutions on economic growth differs across decades, with a coefficient of  $\lambda$  that is not as statistically significant in some decades as it is, for exam-

ment. The sum of each country's sub-category scores translates to a rating from 1 to 7, with a greater value indicating less freedom.

	(1)	(2)	(3)	(4)		(5)	(6)	(7)	(8)
	β	$\beta^*$	$(eta-eta^*)$	$(\lambda  imes \delta)$		β	$\beta^*$	$(eta-eta^*)$	$(\lambda  imes \delta)$
				Controlli	ng for	Human C	apital		
			HC PWT					HC BI	
2000s	-0.575	-1.359	0.784	0.731		-0.575	-1.190	0.615	0.614
2010s	-0.440	-0.904	0.464	0.449		-0.440	-0.965	0.525	0.508
				Contro	lling fo	or Instituti	ions		
			Democracy	/				Political Righ	ts Index
2000s	-0.629	-0.686	0.057	0.057		-0.570	-0.661	0.091	0.092
2010s	-0.436	-0.497	0.061	0.069		-0.416	-0.559	0.143	0.142

**Table 7**: Decomposition of  $\beta$  : The role of institutions

ple, in the 2000s, when we observe absolute convergence in income levels. Conversely, columns (4) and (8) show that  $\delta$  is positive and statistically significant at the 1 per cent level in all periods, suggesting that institutions are highly related to the level of GDP.

To compare the contribution of each variable to the process of convergence in GDP in recent decades, Table 7 displays the coefficients of absolute convergence and convergence conditional only to human capital (upper part of the Table), and convergence conditional only to institutions (lower part of the table) for the years 2000s and 2010s. The upper part of the table shows than once we control for human capital, the coefficient of convergence more than doubles. In the decade of the 2000s, the difference between absolute and conditional convergence is 0.784. The convergence coefficient increases from -0.575 when we do not include any control to -1.359 when we control for human capital. In contrast, the lower part of the table shows that the convergence coefficient hardly changes once we control for institutions. For the same period, the difference between unconditional and conditional convergence is 0.057, the coefficient of unconditional convergence is -0.629 and that of convergence conditional on the measure of democracy is -0.686. The difference between absolute and conditional convergence with the political rights index, an increment that is still quite below the increment when we control for human capital.

These results suggest that the role of human capital in the process of convergence in recent decades is not picking up the effect of institutions. We find that human capital has played a stronger role than institutions in the convergence process. While the coefficient of absolute convergence hardly increases when we control for institutions, it more than

doubles when we condition on human capital.

#### 4..3 Fixed effects

Acemoglu and Molina (2021) argue that Kremer et al. (2021) results are driven by the estimation of a model that does not account for unobserved permanent determinants of GDP per capita. The correlation between the unobserved fixed effect and the initial level of income will be positive, as richer countries tend to have persistent factors that promote development. The omission of such factors in a model where the dependent variable is the level of income will bias the coefficient of the initial level of income upwards. In equation (7), where the dependent variable is the growth rate, the convergence coefficient  $\beta$  will be biased downward. In fact, Acemoglu and Molina (2021) showed that the omission of country fixed effects biases the convergence coefficient towards zero. They also show that the bias can be time-varying and accounts for the different patterns of convergence over time.

However, controlling for unobserved permanent determinants with the fixed-effect estimator will also lead to biased coefficients in dynamic models in which the lagged dependent variable enters as an explanatory variable (see Nickell, 1981).<sup>14</sup> The coefficient of the initial level of income in a level equation will be biased downward, resulting in an upward bias of  $\beta$ . The bias will be greater the shorter the time period. Moreover, Hauk and Wacziarg (2009) state that the higher speed of convergence with the FE estimator found in the literature is likely to be due to the exacerbation of the measurement error bias. They use simulation methods to assess the bias properties of commonly used estimators in the presence of measurement error and show that the fixed-effects estimator overestimates the speed of convergence, while applying OLS on variables averaged over time provides a more accurate estimate.

As Barro (2015) points out, there is a trade-off between two types of biases: the Nickell bias in the fixed effect estimator, that biases  $\beta$  upwards, and the omitted variable bias in the OLS estimator, that results in a downward bias of  $\beta$ .

<sup>&</sup>lt;sup>14</sup> By subtracting the mean to eliminate the individual effects, the fixed effect estimator will be biased since the transformed lagged dependent variable and the transformed error term will be correlated. The resulting correlation creates a downward bias in the estimate of the coefficient of the lagged dependent variable. <sup>14</sup> Hauk and Wacziarg (2009) also find that the GMM estimator of Arellano and Bond (1991), which deals with the fixed effects by taking first differences and using instrumental variables, has a problem of weak instruments in small samples, biasing the GMM estimates toward their fixed-effects counterparts

					Fixed Effe	ects			
	(1)	(2)		(3)	(4)	(5)	 (6)	(7)	(8)
					HC PWT			HC BL	
	β	β	-	$\beta^*$	λ	δ	$\beta^*$	λ	δ
1960s	-1.464 <sup>a</sup>	-1.314 <sup>a</sup>		-1.297 <sup>a</sup>	0.722	0.009	-1.078 <sup>a</sup>	-0.775	0.005
	(0.304)	(0.269)		(0.348)	(1.215)	(0.011)	(0.335)	(1.244)	(0.010)
1970s	$-1.465^{a}$	-1.376 <sup>a</sup>		$-1.470^{a}$	0.914	0.003	-1.403	0.191	-0.002
	(0.254)	(0.225)		(0.251)	(0.966)	(0.009)	(0.248)	(1.034)	(0.009)
1980s	$-1.884^{a}$	$-1.748^{a}$		$-2.347^{a}$	3.899 <sup>a</sup>	0.012	-2.220 <sup>a</sup>	3.251 <sup>a</sup>	-0.001
	(0.250)	(0.221)		(0.267)	(1.046)	(0.009)	(0.263)	(1.140)	(0.008)
1990s	-1.775 <sup>a</sup>	$-1.697^{a}$		-1.595 <sup>a</sup>	0.130	$0.014^{c}$	$-1.397^{a}$	-1.503	-0.003
	(0.245)	(0.218)		(0.305)	(1.238)	(0.008)	(0.299)	(1.354)	(0.008)
2000s	$-2.522^{a}$	-2.288 <sup>a</sup>		$-3.090^{a}$	4.891 <sup>a</sup>	0.007	-2.882 <sup>a</sup>	3.780 <sup>a</sup>	-0.006
	(0.234)	(0.209)		(0.285)	(1.241)	(0.008)	(0.278)	(1.337)	(0.008)
2010s	$-2.285^{a}$	-2.328 <sup>a</sup>		$-2.994^{a}$	4.208 <sup>a</sup>	0.001	-2.859 <sup>a</sup>	3.591 <sup>a</sup>	$-0.018^{b}$
	(0.241)	(0.218)		(0.331)	(1.426)	(0.008)	(0.323)	(1.600)	(0.008)
Constant	$1.458^{a}$	$1.540^{a}$		$1.665^{a}$	1.665 <sup>a</sup>	0.391 <sup>a</sup>	1.573 <sup>a</sup>	1.573 <sup>a</sup>	0.403 <sup>a</sup>
	(0.330)	(0.291)		(0.291)	(0.291)	(0.011)	(0.290)	(0.290)	(0.011)
$R^2$	0.266	0.436		0.469	0.469	0.851	0.466	0.466	0.872
Obs	821	821		821	821	821	821	821	821
Countries	140	140		140	140	140	140	140	140
Year FE	YES	YES		YES	YES	YES	YES	YES	YES
Country FE	YES	YES		YES	YES	YES	YES	YES	YES
Dummy outliers	NO	YES		YES	YES	YES	YES	YES	YES

**Table 8**: Conditional *β*-convergence with fixed effects

#### 4..3.1 Fixed-effects Estimator

We check the robustness of the results when we control for unobserved permanent determinants of the growth rates. Table 8 shows the results when we estimated equations (7-9) with the fixed effects estimator (FE). The results in columns (1)-(2) show that, conditional to specific characteristics of countries, there is convergence in all periods and  $\beta$  is negative and statistically significant in all decades. As expected, the convergence coefficient with FE is much higher than that with the OLS estimator. This is partly due to the role of time invariant characteristics that are specific to each country, and partly due to the upward bias associated to the FE estimator in dynamic models. The results hold with both measures of human capital.

Columns (3) and (6) show that, when we control for human capital, the convergence coefficient increases, suggesting that even controlling for fixed effects, human capital has played a role in the convergence process. Columns (5) and (8) also indicate that, while

human capital is an important determinant of growth rates, there is no evidence that GDP per capita has influenced human capital because the decade coefficients of GDP in equation (9) are not statistically significant. This result minimizes the problem of endogeneity in the growth equation, as more development is not associated with more human capital in the fixed effects model.

In Table 9 we compare the convergence coefficient in different scenarios. It is important to note that, when we include fixed effects, we are controlling for permanent specific characteristics of countries that are difficult to observe or measure. The convergence coefficient in this specification will be convergence conditional to the invariant characteristics of countries and should not be treated as absolute convergence. Thus, we compare the estimated beta coefficients when we do not control for fixed effects ( $\beta^{OLS}$ ), when we condition on time-invariant country-specific characteristics with the FE estimator ( $\beta^{FE}$ ), when we control for fixed effects and human capital ( $\beta^{FE}_H$ ), and when we control for fixed effects and institutions ( $\beta^{FE}_{IN}$ ).

The results in the upper part of the table show that the convergence coefficient in the decade of the 2000s increases by 1.713 points in the fixed effects model ( $\beta^{OLS} - \beta^{FE}$ ). As stated before, this huge increment is associated with both the role of permanent country-specific characteristics, and the upward bias associated with the fixed effects. The role of human capital conditioning on fixed effects ( $\beta^{FE} - \beta_H^{FE}$ ) is shown in column (5). Once we control for human capital the convergence coefficient increases by 0.802 points with the human capital measure from the PWT. Interestingly, this increment is similar to that found in the OLS model (0.784). Although the increment is similar in absolute terms to the one in the OLS model, the interpretation is different. In the OLS model we compare absolute convergence with convergence conditional to human capital. In the fixed-effect model, we asses how convergence, conditional to time-invariant characteristics, increases once we additionally control for human capital. In the decade of the 2010s, the increment in the convergence coefficient once we control for human capital is 0.667.

The lower part of the table shows the results when we control for institutions instead of for human capital. In the fixed effects model, the role of institutions in the process of convergence is in line with previous findings. For the period 2000-2010 the coefficient of convergence hardly changes when we control for the measure of democracy ( $\beta^{FE} - \beta_{IN}^{FE} = -0.016$ ), and increases by 0.179 points when we measure institutions with the Political Rights Index. Similar results are found for the 2010s decade.

<sup>&</sup>lt;sup>14</sup> We also show the convergence coefficient in each case as the number of observations is different.

	(10)		$\beta^{FE} - \beta^{FE}_H$		0.594	0.531			$\beta^{FE} - \beta^{FE}_{IN}$	-0.179	-0.107
	(6)		$\beta^{OLS} - \beta^{FE}$	BL	1.713	1.888		ghts Index	$\beta^{OLS} - \beta^{FE}$	4.128	4.401
	(8)		$\beta_{H}^{FE}$	HC	-2.882	-2.859		Political Ri	$\beta_{\rm IN}^{\rm FE}$	-4.519	-4.710
ысе	(2)	oital	$\beta^{FE}$		-2.288	-2.328		I	$\beta^{FE}$	-4.698	-4.817
– converge	(9)	luman Caj	$\beta_{OTS}$		-0.575	-0.440	tions		β <sub>OLS</sub>	-0.570	-0.416
omposition of $\beta$ - <i>Fixed Effects</i>	(5)	Controlling for H	$\beta^{FE} - \beta^{FE}_H$		0.802	0.667	trolling for Institu		$\beta^{FE} - \beta^{FE}_{IN}$	-0.016	-0.005
Table 7: Dec	(4)		$\beta^{OLS} - \beta^{FE}$	PWT	1.713	1.888	Con	ocracy	$\beta^{OLS} - \beta^{FE}$	2.384	2.588
	(3)		$\beta_{H}^{FE}$	HC I	-3.090	-2.994		Demo	$\beta_{\rm IN}^{\rm FE}$	-2.997	-3.019
I	(2)		$\beta^{FE}$		-2.288	-2.328			$\beta^{FE}$	-3.013	-3.024
	(1)		$\beta_{OTS}$		-0.575	-0.440			BOLS	-0.629	-0.436
					2000s	2010s				2000s	2010s

**Table 9**: Decomposition of  $\beta$  -convergence: human capital, fixed effects and institutions

Overall, these results suggest that the role of human capital in the process of absolute convergence in GDP, observed from the 2000s, is not driven by country-specific characteristics that are omitted in the OLS specification. Even in the presence of fixed effects, the convergence coefficient increases when we control for human capital, and the increment is much higher than when we control for institutions.

#### 4..3.2 Regional dummies

Estimates of the speed of convergence in the influential model of Mankiw et al. (1992) assume initial efficiency to be uncorrelated to the regressors. If that is not the case, the estimates will be biased. A way to deal with initial efficiency is to treat it as a fixed effect, as in Caselli et al. (1996) and Islam (1995). However, as pointed out above, the fixed-effect estimator also comes with its own problems. The Nickel type bias and the exacerbation of measurement error deliver biased and inaccurate estimates.

An alternative approach, suggested by Temple (1998), is to proxy for unobserved initial efficiency by controlling for regional dummies. The idea is that most of the variation in technical efficiency is between regional groups of countries, rather than within them.

In Table 10 we follow this approach and include regional dummies for East Asia and the Pacific, Latin America and the Caribbean, Sub-Saharan African countries, and Advanced economies. Column (1) shows the results of equation (7) adding regional dummies in the set of controls. As expected, the coefficient of the regional dummies is positive for the Advanced economies and East Asia and the Pacific region, and negative for Latin American and Sub-Saharan African countries. If regional dummies are good proxies for initial efficiency, we should interpret the coefficients as conditional convergence, given initial efficiency ( $\beta^{RD}$ ).

When we control for regional dummies, we observe convergence from the 1980s onwards, and a large increment in the convergence coefficients. These results are in the same direction as those when we control for fixed effects, though the increment in the convergence coefficients is lower in quantitative terms. The results in the remaining columns show that even controlling for regional dummies, the convergence coefficient increases when we condition on human capital measures.

Table 11 shows the coefficients of absolute convergence, conditional convergence, and the increment in  $\beta$ -convergence for the 2000s and the 2010s when we condition on human capital and when we condition on institutions. Again, the convergence coefficient increases when we condition on human capital, and slightly increases or even reduces

		Con	trolling fo	r Human	Capital and	l Re	gional D	ummies	
	(1)		(2)	(3)	(4)		(5)	(6)	(7)
				$h_{PWT}$				$h_{BL}$	
	β		$\beta^*$	λ	δ	-	$\beta^*$	λ	δ
1960s	-0.215		$-0.715^{b}$	$3.108^{b}$	0.124 <sup>a</sup>	_	-0.633 <sup>c</sup>	$2.855^{b}$	0.128 <sup>a</sup>
	(0.206)		(0.357)	(1.236)	(0.020)		(0.332)	(1.244)	(0.019)
1970s	-0.164		-0.320	$1.805^{c}$	0.089 <sup>a</sup>		-0.367	$2.126^{b}$	0.090 <sup>a</sup>
	(0.188)		(0.252)	(0.933)	(0.019)		(0.246)	(0.980)	(0.019)
1980s	-0.636 <sup>a</sup>		$-1.334^{a}$	4.862 <sup>a</sup>	$0.105^{a}$		$-1.304^{a}$	5.225 <sup>a</sup>	0.098 <sup>a</sup>
	(0.242)		(0.298)	(0.925)	(0.015)		(0.289)	(1.043)	(0.016)
1990s	$-0.351^{b}$		-0.123	-0.457	0.126 <sup>a</sup>		-0.134	-0.593	0.116 <sup>a</sup>
	(0.147)		(0.255)	(0.989)	(0.013)		(0.250)	(1.125)	(0.014)
2000s	-1.071 <sup>a</sup>		-1.477 <sup>a</sup>	3.149 <sup>a</sup>	$0.104^{a}$		$-1.465^{a}$	3.373 <sup>a</sup>	0.098 <sup>a</sup>
	(0.166)		(0.253)	(0.868)	(0.014)		(0.254)	(0.994)	(0.014)
2010s	$-1.019^{a}$		$-1.202^{a}$	$1.807^{b}$	0.110 <sup>a</sup>		$-1.249^{a}$	2.297 <sup>a</sup>	0.096 <sup>a</sup>
	(0.149)		(0.187)	(0.781)	(0.014)		(0.178)	(0.858)	(0.013)
Advanced	0.819 <sup>a</sup>		0.321	0.321	$0.242^{a}$		$0.456^{b}$	$0.456^{b}$	0.169 <sup>a</sup>
	(0.251)		(0.238)	(0.238)	(0.027)		(0.229)	(0.229)	(0.025)
East Asia Pacific	1.501 <sup>a</sup>		$1.407^{a}$	$1.407^{a}$	0.021		$1.354^{a}$	$1.354^{a}$	$0.038^{c}$
	(0.296)		(0.291)	(0.291)	(0.023)		(0.290)	(0.290)	(0.023)
Latin America	$-0.705^{a}$		$-0.840^{a}$	$-0.840^{a}$	$0.045^{c}$		$-0.810^{a}$	$-0.810^{a}$	0.030
	(0.234)		(0.225)	(0.225)	(0.023)		(0.226)	(0.226)	(0.021)
S.S. Africa	-1.797 <sup>a</sup>		$-1.459^{a}$	$-1.459^{a}$	-0.164 <sup>a</sup>		$-1.455^{a}$	$-1.455^{a}$	-0.149 <sup>a</sup>
	(0.261)		(0.252)	(0.252)	(0.025)		(0.253)	(0.253)	(0.023)
Constant	2.336a		$2.564^{a}$	$2.564^{a}$	$0.409^{a}$		2.499 <sup>a</sup>	2.499 <sup>a</sup>	$0.434^{a}$
	(0.319)		(0.321)	(0.321)	(0.039)		(0.317)	(0.317)	(0.037)
$R^2$	0.455		0.495	0.495	0.698		0.494	0.494	0.696
Obs	821		821	821	821		821	821	821
Year FE	YES		YES	YES	YES		YES	YES	YES
Dummy outliers	YES		YES	YES	YES		YES	YES	YES

**Table 10**: Conditional and Unconditional  $\beta$  – *convergence* 

when we condition on institutions.

#### 5. Conclusions

This paper shows evidence of absolute convergence in human capital indicators in a sample of 140 countries. Unconditional to other factors, countries that start with lower levels of human capital have experienced higher growth rates in education in all decades from 1970 to 2020.

The literature has mainly focused on two types of convergence:  $\sigma$ - convergence and  $\beta$ -convergence. We find both types of convergence in human capital. From the late

	(1)	(2)	(3)	(4)		(5)	(6)	(7)	(8)
	β	$\beta^*$	$(eta-eta^*)$	$(\lambda  imes \delta)$		β	$eta^*$	$(eta-eta^*)$	$(\lambda  imes \delta)$
				Controlling	for Hu	man Capil	tal		
			HC PWT					HC BL	
2000s	-1.071	-1.477	0.406	0.327		-1.071	-1.465	0.394	0.331
2010s	-1.019	-1.202	0.183	0.199		-1.019	-1.249	0.230	0.221
				Controllin	g for Ir	ıstitutions	;		
			Democracy	/			Poli	itical Rights	Index
2000s	-1.089	-1.038	-0.051	-0.010		-1.069	-1.048	-0.021	0.065
2010s	-0.978	-0.927	-0.051	-0.023		-0.989	-1.036	0.047	0.102

**Table 11**: Controlling for Regional Dummies in the decomposition of  $\beta$ -convergence

1970s, there is a decline in the dispersion of human capital across countries, measured as the variance of the log of human capital. We also find an inverse relationship between the initial levels of human capital and its subsequent growth rates. This process of  $\beta$ -convergence in human capital has been increasing over time.

Convergence in human capital has been the result of a speeding-up of the regions at the bottom, a slowdown of the regions at the top, and a relative stability of the regions in the middle. Countries in South-Asia, for example, were ranked among those with the lowest levels of human capital in 1960. In 1980, this region started speeding up in human capital investment rates and, from 1990 onward, South Asia is the region with the highest growth rates in human capital. At the other extreme, the advanced economies started with the highest level of education in 1960, but experienced the lowest growth rates in human capital between 2010 and 2020. In the middle, countries in Latin America and the Caribbean have ranked fourth in terms of human capital growth rates in almost all decades.

Recent empirical evidence has shown a change in the slope in the relationship between per capita income and its growth rates from the 2000s, suggesting absolute convergence in per capita income. As convergence in human capital preceded that of per capita income, human capital may have contributed to the process of absolute convergence in per capita income observed in recent years. We use the omitted variable bias formula to test whether human capital has played any role in explaining convergence in per capita income from 2000 to 2019. We find that the coefficient of  $\beta$ – convergence, conditional to the initial level of human capital, is more than double that the coefficient of unconditional convergence. The difference between both coefficients is explained by the role of human capital.

We show that this result is robust to alternative measures of per capita GDP and even holds after we control for country fixed effects. Additionally, this result is not driven by the role of institutions in the convergence process either. In fact, we find a much higher convergence coefficient when we condition on human capital than when we do it on institutions.

Finding absolute convergence in per capita income is a new phenomenon among countries with distinct levels of development. We show that in a sample that represents most of the countries in the world, the patterns of human capital convergence have helped to account for the observed patterns of income convergence in the recent decades. This evidence leads to the suggestion that human capital is not only an important determinant of growth rates, but can also increase the speed of convergence and close the gap in GDP per capita between poor and rich economies.

#### References

- Acemoglu, D., F. A. Gallego and J.A. Robinson (2014): "Institutions, human capital, and development." Annual Review of Economics, 6(1), 875-912.
- Acemoglu, D., S. Johnson and J. A, Robinson (2005): "Institutions as a fundamental cause of long-run growth," in P. Aghion and S. N. Durlauf (eds.), *Handbook of Economic Growth*, 1, 385-472.
- Acemoglu, D., and C. A. Molina (2021): "Converging to Converge? A Comment." National Bureau of Economic Research, WP No. w28992.
- Acemoglu, D.; S. Naidu, P. Restrepo and J.A Robinson (2019): "Democracy does cause growth." Journal of Political Economy, 127 (1), 47(100).
- Arellano, M. and S. Bond (1991): "Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations", *Review of Economic Studies*, vol. 58(2), pp. 277–97.
- Barro, R. J. (1991): "Economic growth in a cross section of countries." *The Quarterly Journal of Economics* 106 (2), 407-443.
- Barro, R. J. (2015): "Convergence and Modernisation." Economic Journal, 125 (585), 911-42.
- Barro, R. and J.W. Lee (2013): "A New Dataset of Educational Attainment in the World, 1950-2010." Journal of Development Economics, 104, 184–198.
- Barro, R. J., and X, Sala-i-Martin (1992): "Convergence." Journal of Political Economy, 100(2), 223-251.
- Baumol, W.J. (1986). "Productivity growth, convergence, and welfare: what the long-run data show." *American Economic Review*, 76(5), 1072–85.
- Caselli, F. (2005): "Accounting for Cross-Country Income Differences." *Handbook of Economic Growth*, 1(A), pp 679-741.
- Caselli, F., G. Esquivel, and F. Lefort (1996): "Reopening the Convergence Debate: a New Look at Cross-Country Growth Empirics." *Journal of Economic Growth*, 1 (3), 363-389.
- Cohen, D. and L. Leker (2014): "Health and Education: Another Look with the Proper Data", mimeo Paris School of Economics.
- Cohen, D. and M. Soto (2007): "Growth and Human Capital: Good Data, Good Results." *Journal of Economic Growth*, 12(1), 51-76.
- de la Fuente, A. and R. Doménech (2006): "Human Capital in Growth Regressions: How Much Difference Does Data Quality Make." *Journal of the European Economic Association*, 4(1), 1-36.
- de la Fuente, A. and R. Doménech (2006): "Cross-country data on skills and the quality of schooling: A selective survey." *Journal of Economic Surveys*, (forthcoming).
- De Long, J.B. (1988). "Productivity growth, convergence, and welfare: comment", *American Economic Review*, 78(5), 1138–54.
- Durlauf, S. N., P. A. Johnson, and J. R. Temple (2005): "Growth Econometrics," in P Aghion and S. N. Durlauf (eds.), *Handbook of Economic Growth* 1, 555-677.
- Feenstra, R. C., R. Inklaar, and M. P. Timmer (2015): "The Next Generation of the Penn World Table," American Economic Review, 105 (10), 3150–3182.
- Glaeser, E. L., R. La Porta, R., F. Lopez-de-Silanes and A. Shleifer (2004): "Do institutions cause growth?." *Journal of economic Growth*, 9(3), 271-303.
- Hall, R. E. and C. I. Jones (1999): "Why Do Some Countries Produce So Much More Output Per Worker Than Others?" *The Quarterly Journal of Economics*, 114 (1), pp. 83-116.
- Hauk, W.R. and Wacziarg, R. (2009): "A Monte Carlo study of growth regressions", *Journal of Economic Growth*, vol. 14(2), pp. 103–47.
- Islam, N. (1995): "Growth Empirics: A Panel Data Approach." *Quarterly Journal of Economics*, 110 (4), 1127–70.
- Islam, N. (2003). "What have we learned from the convergence debate?". Journal of Economic Surveys, 17, 309–362.

- Johnson, S., W. Larson, C. Papageorgiou, and A. Subramanian (2013): "Is Newer Better? Penn World Table Revisions and Their Impact on Growth Estimates," *Journal of Monetary Economics*, 60 (2), 255–274.
- Johnson, P. and C. Papageorgiou (2020): "What Remains of Cross-Country Convergence?" JJournal of Economic Literature, 58 (1), 129-75.
- Klenow, P., Rodriguez-Clare, A. (1997). "The neoclassical revival in growth economics: has it gone too far?". In: Bernanke, B., Rotemberg, J. (Eds.), Macroeconomics Annual 1997. MIT Press, Cambridge.
- Kremer, M., J. Willis and Y. You (2021): "Converging to Convergence." NBER Macroeconomic Annual, 36.
- Lipset, S. M. (1960): Political Man: The Social Basis of Modern Politics. New York: Doubleday.
- Mankiw, N. G., D. Romer, and D. N. Weil (1992): "A Contribution to the Empirics of Economic Growth." *The quarterly journal of economics*, 107 (2), 407-437.
- Nerlove, M. (2000): "Growth rate convergence: fact or artifact?" in J. Krishnakumar and E. Ronchetti (eds.), *Panel Data Econometrics: Future Directions*, 3–34, Amsterdam: North-Holland.
- Nickell, S. (1981): "Biases in dynamic models with fixed effects", Econometrica, vol. 49(6), pp. 1417-26.
- O'Neill, D. (1995). "Education and Income Growth: Implications for Cross-Country Inequality." *Journal of Political Economy*, 103(6), 1289-1301.
- Patel, D., J. Sandefur, and A. Subramanian (2021): "The New Era of Unconditional Convergence." *Journal of Development Economics*, 152, 102687.
- Pritchett, L. (1997): "Divergence, big time." Journal of Economic Perspectives, 11 (3), 3-17.
- Rodrik, D. (2013): "Unconditional Convergence in Manufacturing." *The Quarterly Journal of Economics* 128 (1), 165-204.
- Sab, R. and S.C. Smith. "Human Capital Convergence: A Joint Estimation Approach." *IMF Staff Papers*, 49(2), 200-211.
- Temple, J. (1998). "Robustness tests of the augmented Solow model." *Journal of Applied Econometrics*, 13 (4), 361–375.
- Young, A. T., M. J. Higgins, and D. Levy (2008): "Sigma convergence versus beta convergence: Evidence from us county-level data." *Journal of Money, Credit and Banking*, 40 (5), 1083-1093.

#### **Creating Opportunities**



## **Working Papers**

#### 2022

22/12 **Amparo Castelló-Climent and Rafael Doménech:** Converging to Convergence: The Role of Human Capital.

22/11 Gül Yücel, Mert Zobi: Positioning of Türkiye in Global Value Chains (GVCs).

22/10 **Ángel de la Fuente:** La evolución de la financiación de las comunidades autónomas del régimen común, 2002-2022.

22/09 Luis Antonio Espinosa Carrasco, Juan José Li Ng: México | ¿El confinamiento por COVID-19 redujo la contaminación del aire?

22/08 **Ángel de la Fuente:** La liquidación de 2020 del sistema de financiación de las comunidades autónomas de régimen común.

22/07 **Ángel de la Fuente y Pep Ruiz Aguirre:** Series largas de VAB y empleo regional por sectores, 1955-2020. Actualización de RegData-Sect hasta 2020.

22/06 Guillermo Jr. Cárdenas Salgado, Juan José Li Ng, Héctor Ortega Rosas, Susana Ramos Villaseñor, Carlos Serrano Herrera, Elmer Solano Flores: ¿Los SMS nudges promueven la salud financiera?

22/05 **Shulin Shen, Le Xia, Yulin Shuai, Da Gao:** Measuring news media sentiment using Big Data for Chinese stock markets.

22/04 Ángel de la Fuente: Las finanzas autonómicas en 2021 y entre 2003 y 2021.

22/03 **José E. Boscá, José Cano, Javier Ferri:** Covid-19 in Spain during 2021: what have vaccines achieved and what is the health cost of vaccine hesitancy?.

22/02 Adriana Haring and Mario Iparraguirre: Argentina. Futuro de los sectores post pandemia.

22/01 **Ángel de la Fuente:** Series largas de algunos agregados económicos y demográficos regionales: actualización de RegData hasta 2020 (RegData y RegData Dem versión 6.1-2020).

#### 2021

21/11 **Ángel de la Fuente and Rafael Doménech:** Cross-country data on skills and the quality of schooling: a selective survey.

21/10 **Ángel de la Fuente:** La evolución de la financiación de las comunidades autónomas de régimen común, 2002-2019.

21/09 **Ángel de la Fuente:** La liquidación de 2019 del sistema de financiación de las comunidades autónomas de régimen común.

21/08 Rodolfo Méndez-Marcano: A global vector autoregressive model for banking stress testing.



21/07 Ali B. Barlas, Seda Guler Mert, Berk Orkun Isa, Alvaro Ortiz, Tomasa Rodrigo, Baris Soybilgen and Ege Yazgan: Big Data Information and Nowcasting: Consumption and Investment from Bank Transactions in Turkey.

21/06 **Ángel de la Fuente y Rafael Doménech:** El nivel educativo de la población en España y sus regiones: actualización hasta 2019.

21/05 Saidé Salazar, Jaime Oliver, Álvaro Ortiz, Tomasa Rodrigo and Ignacio Tamarit: ESP/ Patrones de Consumo de Efectivo vs Tarjeta en México: una aproximación Big Data. ING / Cash Vs Card Consumption Patterns in Mexico: A Machine Learning Approach.

21/04 **Ángel de la Fuente:** La financiación autonómica en 2020: una primera aproximación y una propuesta de cara a 2021.

21/03 Ángel de la Fuente: Las finanzas autonómicas en 2020 y entre 2003 y 2020.

21/02 **Joxe Mari Barrutiabengoa, J. Julián Cubero and Rodolfo Méndez-Marcano:** Output-side GHG Emissions Intensity: A consistent international indicator.

21/01 **Ángel de la Fuente y Pep Ruiz:** Series largas de VAB y empleo regional por sectores, 1955-2019 Actualización de *RegData-Sect* hasta 2019.

CLICK TO ACCESS THE WORKING PAPERS PUBLISHED IN Spanish and English



### DISCLAIMER

The present document does not constitute an "Investment Recommendation", as defined in Regulation (EU) No 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse ("MAR"). In particular, this document does not constitute "Investment Research" nor "Marketing Material", for the purposes of article 36 of the Regulation (EU) 2017/565 of 25 April 2016 supplementing Directive 2014/65/EU of the European Parliament and of the Council as regards organisational requirements and operating conditions for investment firms and defined terms for the purposes of that Directive (MIFID II).

Readers should be aware that under no circumstances should they base their investment decisions on the information contained in this document. Those persons or entities offering investment products to these potential investors are legally required to provide the information needed for them to take an appropriate investment decision.

This document has been prepared by BBVA Research Department. It is provided for information purposes only and expresses data or opinions regarding the date of issue of the report, prepared by BBVA or obtained from or based on sources we consider to be reliable, and have not been independently verified by BBVA. Therefore, BBVA offers no warranty, either express or implicit, regarding its accuracy, integrity or correctness.

This document and its contents are subject to changes without prior notice depending on variables such as the economic context or market fluctuations. BBVA is not responsible for updating these contents or for giving notice of such changes.

BBVA accepts no liability for any loss, direct or indirect, that may result from the use of this document or its contents.

This document and its contents do not constitute an offer, invitation or solicitation to purchase, divest or enter into any interest in financial assets or instruments. Neither shall this document nor its contents form the basis of any contract, commitment or decision of any kind.

The content of this document is protected by intellectual property laws. Reproduction, transformation, distribution, public communication, making available, extraction, reuse, forwarding or use of any nature by any means or process is prohibited, except in cases where it is legally permitted or expressly authorised by BBVA on its website www.bbvaresearch.com.